Author’s response to reviews


Authors:

Subramanian Vaidyanathan (vaidyanathansiu@hotmail.com)
Paul Mansour (Paul.Mansour@mail.soh-tr.nwest.nhs.uk)
Munehisa Ueno (camtmune@saitama-med.ac.jp)
Kazuto Yamazaki (KazutoYamazaki@aol.com)
Meenu Wadhwa (mwadhwa@nibsc.ac.uk)
Bakul M Soni (Bakul.Soni@mail.soh-tr.nwest.nhs.uk)
Gurpreet Singh (Gurpreet.Singh@mail.soh-tr.nwest.nhs.uk)
Peter L Hughes (Peter.Hughes@mail.soh-tr.nwest.nhs.uk)
Ian D Watson (ian.watson@mail.soh-tr.nwest.nhs.uk)
Pradipkumar Sett (Pradipkumar.Sett@mail.soh-tr.nwest.nhs.uk)

Version: 10 Date: 26 Apr 2002

Dear Editor
I included a statement under "Acknowledgement" that the patient’s father gave consent for publication of his son’s case in BioMed Central Urology. I made a statement that the patient, while he was alive, gave permission to send his biopsy specimens to Japan for immunohistochemistry for G-CSF and PTHrP.
Many thanks for considering this manuscript for publication in your esteemed journal.
Yours sincerely
Dr S Vaidyanathan